
Daily Status 4/30/17
Weather:   windy and rainy this morning. May clear up in the afternoon.

Sensor Status Overview
   A number of sonics, ~7, were down this morning. It appears rain was the reason. Most are back up.
  Three Licors were down but I was able to get running again via PIO commands.
  TRH 10m and 60m on rsw06 were down due to fan current going to high. I was able to get them back
       up via PIO. The max current cutoff has been increased, will monitor.
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